UC Berkeley UC Berkeley Previously Published Works

Title

Classic Selective Sweeps Revealed by Massive Sequencing in Cattle

Permalink

https://escholarship.org/uc/item/9nb6t8cq

Journal

PLOS Genetics, 10(2)

ISSN

1553-7390

Authors

Qanbari, Saber Pausch, Hubert Jansen, Sandra <u>et al.</u>

Publication Date

2014

DOI

10.1371/journal.pgen.1004148

Copyright Information

This work is made available under the terms of a Creative Commons Attribution License, available at <u>https://creativecommons.org/licenses/by/4.0/</u>

Peer reviewed

Classic Selective Sweeps Revealed by Massive Sequencing in Cattle

Saber Qanbari^{1*}, Hubert Pausch², Sandra Jansen², Mehmet Somel^{3¤}, Tim M. Strom⁴, Ruedi Fries², Rasmus Nielsen³, Henner Simianer¹

1 Animal Breeding and Genetics Group, Department of Animal Sciences, Georg-August University, Goettingen, Germany, 2 Chair of Animal Breeding, Technische Universitaet Muenchen, Munich, Germany, 3 Departments of Integrative Biology and Statistics, University of California at Berkeley, Berkeley, California, United States of America, 4 Institute of Human Genetics, Helmholtz Zentrum München, Munich, Germany

Abstract

Human driven selection during domestication and subsequent breed formation has likely left detectable signatures within the genome of modern cattle. The elucidation of these signatures of selection is of interest from the perspective of evolutionary biology, and for identifying domestication-related genes that ultimately may help to further genetically improve this economically important animal. To this end, we employed a panel of more than 15 million autosomal SNPs identified from re-sequencing of 43 Fleckvieh animals. We mainly applied two somewhat complementary statistics, the integrated Haplotype Homozygosity Score (iHS) reflecting primarily ongoing selection, and the Composite of Likelihood Ratio (CLR) having the most power to detect completed selection after fixation of the advantageous allele. We find 106 candidate selection regions, many of which are harboring genes related to phenotypes relevant in domestication, such as coat coloring pattern, neurobehavioral functioning and sensory perception including KIT, MITF, MC1R, NRG4, Erbb4, TMEM132D and TAS2R16, among others. To further investigate the relationship between genes with signatures of selection and genes identified in QTL mapping studies, we use a sample of 3062 animals to perform four genome-wide association analyses using appearance traits, body size and somatic cell count. We show that regions associated with coat coloring significantly (P<0.0001) overlap with the candidate selection regions, suggesting that the selection signals we identify are associated with traits known to be affected by selection during domestication. Results also provide further evidence regarding the complexity of the genetics underlying coat coloring in cattle. This study illustrates the potential of population genetic approaches for identifying genomic regions affecting domestication-related phenotypes and further helps to identify specific regions targeted by selection during speciation, domestication and breed formation of cattle. We also show that Linkage Disequilibrium (LD) decays in cattle at a much faster rate than previously thought.

Citation: Qanbari S, Pausch H, Jansen S, Somel M, Strom TM, et al. (2014) Classic Selective Sweeps Revealed by Massive Sequencing in Cattle. PLoS Genet 10(2): e1004148. doi:10.1371/journal.pgen.1004148

Editor: Jonathan K. Pritchard, Stanford University, United States of America

Received August 21, 2013; Accepted December 14, 2013; Published February 27, 2014

Copyright: © 2014 Qanbari et al. This is an open-access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Funding: This research was funded by the German Federal Ministry of Education and Research (BMBF) within the AgroClustEr "Synbreed - Synergistic plant and animal breeding" (FKZ 0315528). The authors acknowledge support by the German Research Foundation and the Open Access Publication Funds of the Göttingen University. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing Interests: The authors have declared that no competing interests exist.

* E-mail: sqanbar@gwdg.de

¤ Current address: Middle East Technical University, Department of Biological Sciences, Ankara, Turkey

Introduction

The available genetic and archaeological evidences date cattle domestication back to the Neolithic period, around 10,000 BCE [1,2]. Modern cattle are thought to have originated from multiple independent domestication events of aurochs (B. primigenius) primarily in southwest Asia and south Asia, resulting in the humpless taurine (B. taurus) and the humped zebu (B. indicus) groups respectively [3,4]. Domestication of cattle had a major impact on human civilization as they provided physical power in agriculture and were a major source of milk, meat and leather products.

Domestication of cattle provides an excellent model of animal evolution. During the domestication process, cattle have adapted in morphology, physiology and behavior to captive life, and have been subject to artificial selection imposed by humans to increase yield, fertility and other processes. As a result, more than 900 breeds, each with distinct characteristics, have emerged throughout the world [5]. The phenotypes associated with domestication include milk and meat production, fertility, appearance including coat coloration, decreased fearfulness, social motivation, and mild temper [6]. The selection affecting these phenotypes has left detectable signatures of selection within the genome of modern cattle [7].

The signatures of selection in the genome, as a beneficial mutation arises and rapidly increases in frequency in the population, can be detected as (i) reduced local variability, (ii) deviations in the Site Frequency Spectrum (SFS) and (iii) increased linkage disequilibrium and extended haplotype structure. These signatures can be used to screen a genome for genes involved in recent adaptation. Numerous statistics have been developed aiming at detecting selection [8,9,10,11,12 and 13].

Previous genome-wide studies to detect positive selection in cattle have used SNP arrays, which suffer from ascertainment biases caused by the process used to discover SNPs [14,15,16] and limited resolution [17,18,19,20,21 and 22]. In addition, these

Author Summary

Domestication of cattle had a major impact on human civilization by providing protein and physical power for agrarian life style. Domestication followed by breed formation has likely left detectable signatures within the genome of modern cattle. Current cattle breeds, for instance, have a more uniform appearance and milder temper than their wild ancestors. The elucidation of these signatures of selection is of interest to identify domestication-related genes that help to genetically improve this economically important species. The development of novel sequencing technologies has enabled higher-resolution genomic analyses of past selection. In this paper, we exploited whole genome sequencing along with multiple statistical metrics to identify regions/genes putatively targeted by selection. We show strong signals of selection near to several candidate genes related to domesticated phenotypes such as coat coloring, neurobehavioral functioning and sensory perception, including KIT, MITF, MC1R, NRG4, Erbb4, TMEM132D and TAS2R16. By means of association mapping we additionally show that candidate selection regions for appearance traits overlap with major coat color QTLs. Our study demonstrates the utility of population based techniques for detecting past selection and is the first attempt to localize signatures of past selection in cattle based on massive re-sequencing of the entire genome.

studies have focused on a single selection signature statistic that typically only detects selection during a certain time in the past. Many selection signals may, therefore, have remained un-detected by previous studies.

In this study, we use whole-genome re-sequencing data of 43 Fleckvieh animals [23], a German dual purpose cattle breed. We apply two different statistics, the integrated Haplotype Homozygosity Score (iHS) [12] and the Composite of Likelihood Ratio (CLR) [11] to detect past selection. iHS finds maximal power when a selected allele segregates at intermediate frequencies in the population, whereas the CLR statistics has most power right after the selected allele has gone to fixation. For this reason, the two statistics are complementary in the type of selection that they detect. Using whole genome sequence information rather than genotypes for pre-selected SNP panels avoids the problems caused by ascertainment. This design thus, provides additional power to detect selection missed by previous studies. In addition, we conduct a Genome-Wide Association mapping Study (GWAS) on appearance traits. We find evidence of strong signatures of selection in cattle during speciation, domestication and breed formation exemplified by several striking selective sweeps colocalized with major QTLs.

We also use the direct sequencing data to examine the pattern of Linkage Disequilibrium (LD) in the Fleckvieh breed. A detailed profile of LD over the entire genome is a quantity of interest, especially for the use in breeding programs implementing genomic selection. Previous studies of LD structure in cattle populations have used low resolution panels of ascertained SNPs mainly selected based on their minor allele frequency (MAF) and position on the genome [24,25, among others].

Results and Discussion

Allele frequency distribution and LD

The data from Jansen et al [23] analyzed in this study includes roughly 20 times more SNPs than the 700K array previously used in cattle for examining LD. The increased SNP density provides a greater coverage of rare and low frequency SNPs than in any previous study based on SNP chip data.

The distribution of allele frequencies follow the same pattern as that observed for high-quality data in many other organisms including human populations [26,27]. As predicted by population genetics theory, the frequency spectrum is a decreasing function, and among rare alleles there is a slight excess in the proportion of non-synonymous mutations relative to intergenic or synonymous mutations (Figure 1). The relative excess of non-synonymous mutations among rare alleles is presumed to be caused by selection acting on slightly deleterious mutations [26,27].

We found a mean value of $r^2 = 0.25$ (sample SD = 0.29) for SNPs less than 20 kb apart (Figure 2). Table S1 summarizes more properties of LD as a function of physical distance. It is evident that average LD does not extend beyond the inter-marker space of 100 Kb across the genome. Previous studies in cattle however, found strong LD extending over several Megabasepairs [24,25, among others]. However, LD as measured by r^2 depends on allele frequencies [28,25]; and the difference between this study and previous studies may partially be explained by the biased SNPs selection on the Illumina Bovine arrays, where SNPs mainly were ascertained based on allele frequency and a uniform distribution over the genome. Additionally, differences in the sample composition may explain the results, as LD is strongly affected by population structure. Characterizing LD using structured populations leads to an inflation of the LD statistics, which might have affected previous studies. Finally, genotyping error reduces apparent LD, and is a major concern for low- and intermediatedepth coverage re-sequencing data.

To test the degree to which the differences in the LD curves are caused by these factors we examined LD in different scenarios. First we plotted the LD curve after excluding low-frequency variants from sequence data. As shown in Figure S1A, LD persists at a higher level when being estimated with frequent alleles. Second, we sub-selected the sequence data for those SNPs present in the 700k chip and compared LD from this sub-set with sequence and array based LD (Figure S1B). In the new dataset LD decayed faster than original array-based LD possibly due to the different sample composition. However, LD persisted at higher levels compared to the sequence LD due to the different allelic profile (p < 0.001). Table S2 summarizes testing the strength of LD in two datasets estimated from SNP pairs in inter-marker distance bins up to 500 Kb. Further, to get insight into the quality of genotypes the concordance rate between sequences vs. arrayderived genotypes was evaluated. We observed a significant concordance of 96.9% (±3.4%) based on 38,246 SNPs tested on chromosome 1. The LD curves before and after filtering out all but the highest quality SNPs were overlapped due to the fairly low discordance rate (data not shown). These results demonstrate that LD in cattle decays at a rate much faster than previously thought.

LD-based estimations of past effective population size [7,25] should be revisited in light of the finding that the 'true' sequencebased LD profile is poorly estimated by SNP-chip based LDestimates. Considering the relatively small effective population size in cattle, population level of LD is unexpectedly low, which suggests that effective population size was considerably larger in the very recent past.

Localizing selective sweeps

Evidence of positive selection was investigated through multiple statistics designed to detect signatures of selective sweeps. We calculated iHS per site and averaged them in non-overlapping 40 Kb windows across the genome, resulting in a total of 62,196



Figure 1. Comparison of the site frequency spectra from resequencing of 43 German Fleckvieh animals. SFS is represented for non-synonymous (Non_syn), synonymous (Syn) and inter-genic polymorphic (Int_genic) variants. doi:10.1371/journal.pgen.1004148.g001



Figure 2. A schematic representation of LD plotted as a function of distance. The decay of LD estimated from bovine SNP arrays of 50K and 700K (n = 1,293) are compared with the sequence data (n = 43). The inner plot displays a higher resolution of LD in pair-wise distances of <250 Kb from sequence data of which r^2 values are down-sampled from all pairwise estimates (for more details see Material and Methods). doi:10.1371/journal.pgen.1004148.g002

windows (Figure 3). The CLR was estimated using an identical grid size across the genome. We focused the analyses on windows for which the values of the statistics fell in the 99th percentile. Respectively, 68 and 73 candidate regions were identified from

iHS and CLR analyses (Tables S3 and S4). There is a substantial overlap between the list of genes identified by CLR and iHS, reflecting the fact that the two tests take advantage of different but correlated patterns of a selective sweep. However, there are also



Figure 3. Genome-wide visualization of selection candidates (top 1% signals) localized by |iHS| (A) and CLR (B) metrics. Each dot represents a non-overlapping window of 40 Kb along BTA1 to BTA29. Panels C and D show a high resolution illustration of the candidate regions for the *KIT/KDR* and *MITF* genes, respectively on BTA6 and BTA22. |iHS| is plotted in overlapping windows of 40 Kb in steps of 5 Kb, and a grid size of 5 Kb was chosen for the CLR statistic. Finally, Panel E is a haplotype bifurcation plot of the *KIT/KDR* genes. doi:10.1371/journal.pgen.1004148.g003

regions solely identified by either metric, possibly because these statistics identify selection acting at different time scales. Two examples of candidate genes are shown in Figure 3: *KIT* and *MITF*, two pigmentation genes on Bos taurus (BTA) chromosomes 6 and 22.

Putatively selected genes

Tables S3 and S4 summarize statistics for the genomic regions harboring the strongest selection signals. We used DAVID [29] to perform a functional analysis based on the list of all genes in the regions showing signatures of a selective sweep. We found no overall significant enrichment of any particular biological process after correction for multiple testing (data not shown). Nevertheless, we note that genes associated with a number of processes previously implicated in domestication-related changes are present within these regions. These include pigmentation, sensory perceptions, brain and neural system (for review see [6]) along with genes of immunity and blood clotting systems. For clarity and based on a priori interest, we divided genes into functional groups in line with domestication-related changes and discuss each group under separate heading. However, as most genes have pleiotropic effects, selection may possibly act on other functional effects of the genes than those highlighted here. In the following sections, we highlight some results from these analyses.

Patterned pigmentation. In mammals, coat color loci influence the development, differentiation, proliferation, and migration of melanocytes, the construction and transport of melanosomes, as well as the synthesis of melanin. In the genomewide screen, the window with the strongest signal ($P_{iHS} = 0.00029$, $P_{CLR} = 0.00011$) coincides with a cluster of tyrosine kinase receptor genes (PDGFRA, KIT and KDR) on BTA6 (Figure 3). Among them, KIT is a widely studied gene with an important role in several critical pathways including melanogenesis. Genetic variation in the KIT gene has been shown to affect coat coloring pattern in a variety of mammals including cattle [30], horses [31], pigs [32] and mice [33]. Another strong selection candidate included microphthalmia-associated transcription factor (MITF, $P_{iHS} = 0.00172$ and $P_{CLR} = 0.00709$) a major candidate for patterned pigmentation in BTA22 [30]. KIT and MITF show complex interactions in that *MITF* is needed for the maintenance of KIT expression in melanoblasts and KIT signaling modulates MITF activity and stability in melanocyte cell lines. The mutual interaction between KIT and MITF is particularly interesting since mutations in any one of them lead to a strikingly overlapping phenotype of early loss of the melanocyte lineage [34].

Another particularly interesting selective sweep candidate in this group overlaps melanocortin 1 receptor (MC1R, $P_{iHS} = 0.00785$ and $P_{CLR} = 0.00156$) on BTA18 (also see Figure S2A), whose permanent activation results in black coat color, whereas loss of function mutations cause red coat color in different mammals including cattle [35]. MC1R gene expression is regulated by the MITF and has an autosomal recessive mode of inheritance (also see GWAS section).

Among the top selection candidates, we noticed three genes of the *NRG– Erbb4* signaling pathway. This pathway is involved in the development and progression of melanocytes [36]. Our results revealed typical hitch-hiked patterns for *NRG4* ($P_{iHS} = 0.00368$, $P_{CLR} = 0.00221$) and *Erbb4* ($P_{iHS} = 0.00203$, $P_{CLR} = 0.00855$) genes along with an extremely deviated SFS for the pro-*NRG2* like locus ($P_{iHS} = 0.66899$, $P_{CLR} = 0.00589$)(see also Figures S3 and S4). Interestingly, the same pathway appears to have been targeted by positive selection in humans [37]. Since precursors to pigment cells are also precursors to nerve cells, variants of genes in this pathway are also reported to be associated with various psychiatric phenotypes [38,39]. Another strong signal that we speculate could be related to coat coloring is the *ULBP3* gene ($P_{iHS} = 0.00691$, $P_{CLR} = 0.00917$) shown to be associated with the "sudden whitening of the hair" phenomenon [40].

Brain development and neurobehavioral functioning. Domesticated species differ from their wild ancestors, notably in behavioral traits such as reduced fear of humans and aggressiveness [6]. We noticed strong signals standing by some genes underlying extreme neurobehavioral phenotypes and psychiatric disorders (for instance see Figures S5, S6 and S7). Among the most pronounced candidates in our list are neuronal genes TMEM132D (P_{iHS} = 0.00087, P_{CLR} = 0.00021), CACNA1C $(P_{iHS} = 0.48366, P_{CLR} = 0.00013), NRXN1 (P_{iHS} = 0.00396,$ $P_{CLR} = 0.00018$) and NPAS3 ($P_{iHS} = 0.34897$, $P_{CLR} = 0.00137$) reported as candidate QTLs for anxiety-related behavior, major depression and high risk of developing schizophrenia [41,42,43,44]. One interesting observation was the presence of GRIK3 (P_{iHS} = 0.00076, P_{CLR} = 0.00761) among the genes with the strongest signal of selection. GRIK3 is a member of glutamine receptors suggested as QTL for reward-related learning [45]. Other putative selection candidate include *OLIG1* ($P_{iHS} = 0.00031$, $P_{CLR} = 0.00016$), LAMC3 ($P_{iHS} = 0.00222$, $P_{CLR} = 0.00325$) and ATL1 ($P_{iHS} = 0.00227$, $P_{CLR} = 0.00557$) genes, which all have central roles in the development of brain cortex and formation of axons. Mutations in the LAMC3 are suggested to cause malformations of occipital cortical development in humans [46]. We speculate that these genes could have been affected by selection targeting at behavioral traits such as a modest temperament during domestication.

Sensory perception. We observed signals for selection targeting several sensory functions including olfaction and taste. The putative sweeps on BTA7 and BTA29 contain four clusters of olfactory receptor (OR) family genes (see Table 1). Olfactory receptors detect and identify a wide range of odors and chemosensory stimuli, a necessity to find food, detect mates and offspring, recognize territories and avoid danger (for review see [47]). OR genes are shown to have been under selection in humans [48] and domesticated animals including dog [49], swine [50] and cattle [7]. They are also reported to be duplicated within the bovine genome [51] suggesting that they may be under strong selection for newly evolving functions.

Another candidate selective sweep was localized within a bitter taste receptor gene (*TAS2R16*, $P_{iHS} = 0.00143$, $P_{CLR} = 0.00887$) that enables animals to properly distinguish food sources and prevent them from ingesting potentially harmful compounds such as noxious defense compounds produced by plants [52]. It has been argued that elimination of the need to search for food in wild animals after their domestication as well as adaptation to new dietary habits may relax the evolutionary constraint acting on these genes [53]. Selection has possibly targeted *TAS2R16* as part of the new dietary habits emerging during cattle domestication.

Immune system and genetic disorders. Infectious diseases have been dominant threats to survival; therefore natural selection is expected to act strongly on innate immunity genes. Among the top selection candidates in our list are *MAGEA13P*-like ($P_{iHS} = 0.00056$, $P_{CLR} = 0.00178$) a member of melanoma-associated antigen family (see for instance Figure S7), *FCRL4* ($P_{iHS} = 0.00141$, $P_{CLR} = 0.00317$) an immunoglobulin receptor gene, *UNC93A* ($P_{iHS} = 0.00220$, $P_{CLR} = 0.08933$), associated with Herpes simplex encephalitis type 1, and finally *TNFRSF9* ($P_{iHS} = 0.00132$, $P_{CLR} = 0.00207$) induced by lymphocyte activation gene. Other noteworthy genes in our list are *CAV1* and *CAV2* ($P_{iHS} = 0.00092$, $P_{CLR} = 0.02540$) involved in Cystic Fibrosis, *DSCAM* ($P_{iHS} = 0.00281$, $P_{CLR} = 0.00005$) implicated in Down

Table 1. A partial list of candidate regions revealed by both iHS and CLR analyses.

#	Chr	Gene ¹	Position(bp) ²	P _{liHS}	P _{CLR}	Function/association	Reference
1	1	OLIG1	1,915,043	0.00031	0.00016	Developing oligodendrocytes	
2	1	DSCAM	142,135,568	0.00281	0.00005	Down syndrome cell adhesion molecule	
3	2	Erbb4	99,904,366	0.00203	0.00855	Pigmentation/Neurobehavioral functioning	[37]
4	3	FCRL4	13,339,357	0.00141	0.00317	Immunoglobulin	
5	3	SLC35D1	78,587,030	0.00061	0.00854	Responsible for skeletal dysplasia	[77]
6	3	GRIK3	109,293,546	0.00076	0.00761	Reward-related learning	[45]
7	4	VWDE	20,025,093	0.00918	0.07143	A carrier of clotting factor VIII (FVIII)	
8	4	CAV1 & CAV2	52,237,367	0.00092	0.02540	Cystic Fibrosis	
9	4	TAS2R16	88,279,974	0.00143	0.00887	Bitter taste receptor, type 2	
10	5	CACNA1C	109,068,036	0.48366	0.00013	Bipolar disorder and schizophrenia	[42]
11	6	KIT/KDR	72,087,087	0.00029	0.00011	Pigmentation	[30]
12	7	OR	15,246,693	0.00502	0.00194	Olfaction perception	
13	7	OR	43,810,382	0.00113	0.00905	Olfactory receptor family cluster 2	
14	7	LOC783452	52,837,137	0.66899	0.00589	Pro-neuregulin-2, membrane-bound isoform-like	
15	8	OR13C8	96,134,318	0.57280	0.00043	Olfactory receptor, family 13, subfamily C, member 8	
16	8	ASTN2	107,909,740	0.00058	0.00269	Astrotactin 2	
17	9	LIN28B	45,672,801	0.00982	0.00166	Cause gigantism and a delay in puberty	[78]
	9	UNC93A	103,514,079	0.0022	0.08933	Herpes simplex encephalitis type 1	
18	10	ATL1	43,656,593	0.00227	0.00557	Formation and growth of axons	
	11	NRXN1	33,156,535	0.00396	0.00018	Major depression	[43]
19	11	LAMC3	101,292,261	0.00222	0.00325	Cause malformations of occipital cortical development	[46]
20	11	ADAMTS13	104,415,591	0.00711	0.00736	Involved in blood clotting	
21	14	MAGEA13PL	42,271,706	0.00056	0.00178	Immune system	
22	14	LOC100335199	42,856,951	0.00034	0.00010	Tescalcin-like with unknown function	
23	15	OR51A7	50,749,853	0.87230	0.00048	Olfactory receptor, family 51, subfamily A, member 7	
24	16	TNFRSF9	46,579,025	0.00132	0.00207	Immune system	
25	17	ULBP3	40,044,391	0.00691	0.00917	"Sudden whitening of the hair" phenomenon	[40]
26	17	TMEM132D	48,769,405	0.00087	0.00021	Neurobehavioral functioning	[41]
27	18	MC1R	14,785,816	0.00785	0.00156	pigmentation	[35]
28	20	STK10	3,808,800	0.00119	0.00051		
29	21	NRG4	32,012,049	0.00368	0.00221	Pigmentation/Neurobehavioral functioning	[37]
30	21	NPAS3	44,609,230	0.34897	0.00137	High risk of developing schizophrenia	[44]
31	22	MITF	32,017,564	0.00172	0.00709	Pigmentation	[30]
32	25	HS3ST4	23,942,521	0.00494	0.03597	Involved in blood clotting	

¹Only best candidate genes are shown.

²Position stands at the middle of the top hitch-hiked window in the candidate region.

doi:10.1371/journal.pgen.1004148.t001

syndrome, *SLC35D1* ($P_{iHS} = 0.00061$, $P_{CLR} = 0.00854$) responsible for skeletal dysplasia and a strong signal on *Tescalin* like gene ($P_{iHS} = 0.00034$, $P_{CLR} = 0.00010$) with unknown function (Figure S8).

Blood coagulation. We found three genes annotated with blood clotting functions in the region of selection signals. The von Willebrand factor D and EGF domains (*VWDE*, $P_{iHS} = 0.00918$, $P_{CLR} = 0.07143$) a carrier of clotting factor VIII (FVIII), *ADAMTS13* ($P_{iHS} = 0.00711$, $P_{CLR} = 0.00736$) that cleaves *VWDE*, and heparan sulfate (glucosamine) 3-O-sulfotransferase 4 (*HS3ST4*, $P_{iHS} = 0.00494$, $P_{CLR} = 0.03597$) implicated in negative regulation of blood coagulation show signals of selection. This

corresponds with some reports on primates along with human data suggesting potential signatures of positive selection for genes involved in blood clotting pathways [54,55]. Further research on this group of genes would be required to address potential adaptation of blood coagulation genes in cattle.

There is a growing number of genome-wide scans for detecting historical positive selection in cattle and other farm animals. Previous studies in cattle however, have used low resolution panels of ascertained SNPs, mostly based on inter-population comparisons of site frequencies [18,19,20,21,22]. The notable candidate genes reported in these studies include *GHR*, *PDGFRA*, *KIT* and *MC1R* in association with body size and morphology traits in

cattle. These signals generally differ from those reported by the Bovine HapMap consortium [7]. The most recent study based on 700k bovine array employed a composite framework that combines P-values from different tests across multiple breeds [56]. While the polymorphism content in both KIT and MC1R regions were under-represented for conducting an efficient selection scan, we found a poor overlap genome wide in comparison to our results. Besides different marker density and populations in both studies, the differences in the statistical approaches used could explain the discrepancy. The suggested statistical tests applied in this study recover selective events from different time periods and/or for different stages of the selective sweep (e.g., CLR vs. iHS). Furthermore, a selective sweep might be specific for one population and may not appear in other populations. Thus, combining results of multiple tests and across populations may mask real signals which could partly explain the low concordance with our single-breed single-test results.

Validating putative sweeps with GWAS

If our candidate regions are in fact enriched for genes affected by selection related to domestication traits, they should overlap with regions identified in QTL mapping studies on these traits. This hypothesis was verified in an exemplary fashion for the complex trait coat color. Coat color in cattle is usually regarded to be a trait controlled by few loci of large effect [30]. There is a high degree of variation in color apparent within Fleckvieh population. Fleckvieh animals are phenotypically characterized by being red, spotted or not, and having white legs and a white head, animals with red head occasionally occur but are considered as a deviation from the breed standard. We used the coat color traits recorded respectively as the proportion of daughters of bulls 'without spotting' and with 'red head' to validate putative sweeps for coloration phenotypes (Figure 4).

We performed a Genome Wide Association Study (GWAS) on 3602 animals for which genotypes of 15,182,131 SNPs were imputed (see Material and Methods). The GWAS revealed eight SNPs with large effect on the 'proportion of daughters without spotting' (Figure 5B1). The SNP with the most significant effect is a non-genic variant in the vicinity of the *MITF* (P = 2.65e-58) gene on BTA22. Another highly significant but non-genic SNP coincides with a selection candidate region (e.g., Figure 3C) covering the *KIT* and *KDR* (P = 2.46e-44) loci. There are also two significant signals, respectively next to endothelin 3 (*EDN3*) (P = 2.42e-36) and an uncharacterized protein at the proximity of the membrane metallo-endopeptidase (*MME*) gene (P = 5.11e-14). *EDN3* plays a significant role during the early development of melanocytes in their response to ultraviolet radiation, and in pathological conditions including melanoma. Severe pigmentation

defects of mutations in *EDN3* in mice, human and chicken are well-described (for review see [57]). In human models, *MME* is expressed at the surface of melanoma cells and are involved in the regulation of melanogenesis [58]. Although these SNPs were the top association signals for each of the identified QTL, distinguishing causal variants from nearby neutral loci may be the most difficult issue, as those variants possibly stay in LD with the actual selected locus that may produce similar signals due to genetic hitch-hiking.

Genomic relationship matrices were built separately for each chromosome and QTL using imputed variants. The phenotypic variation explained by each chromosome/QTL was then estimated with the effects of all chromosomes/QTL fitted simultaneously using GCTA [59]. All together, the imputed variants explained 82.37% of the phenotypic variation (i.e. 94% of the heritability) (Figure S9). Together, the eight identified QTL explained 49.76% of the phenotypic variation (i.e. 56.8% of the heritability).

The GWAS for the 'proportion of daughters with red head' was modeled in the same way. Visualizing of results shows a different genetic control for the red head phenotype when is compared to the spotting. The strongest GWAS revealed signal (P=6.8e-125) was on the *KIT* locus on BTA6 (Figure 5A1), which alone explained 34.81% of the total variation of the trait (Table 2).

Previous GWAS on coat color in cattle have identified polymorphism in *MC1R*, *KIT* and *MITF* [35,60 and 30]. In both GWA analyses in this study, the *KIT* locus is associated, but *MITF*, *EDN3* and *MME* are related only with spotting phenotype. This observation suggests that a different genetic mechanism regulates these phenotypes in the Fleckvieh breed. Recently, Pausch et al. [60] studied the genetics controlling the peculiar pigmentation surrounding the eyes in Fleckvieh animals using the 700K SNP panel. They found a strong association between *MITF* and *KIT* genes with the UV-protective eye area pigmentation in Fleckvieh cattle, but no association with *EDN3* and *MME* genes. Comparing GWAS results from coat spotting, red head, and eye area pigmentation reveals a complex genetic background for the different pigmentation phenotypes in the Fleckvieh population.

To test for an overlap between the selection candidates and the QTL study, we performed two randomization experiments on coat spotting phenotype using either single SNPs or window-based estimates of |iHS| versus P-values from GWAS (see Methods). Both randomizations revealed no single randomized dataset more extreme than the real data.

While there is no association detected for *MC1R* gene in the GWAS, selection signature analyses suggests that there has been a selective sweep at or near the *MC1R* gene on BTA18 (Figure S2A). *MC1R* variants have been shown to alter pigment synthesis in a range of species, as well as coat color spotting in pigs [61]. At least



Figure 4. Fleckvieh animals with different coat coloring phenotypes. (A) without spot, (B) spotted and (C) red head. The figures were kindly supplied by BAYERN-GENETIK GmbH (http://www.fleckvieh.de). doi:10.1371/journal.pgen.1004148.g004



Figure 5. The visualization of the signals revealed by association analyses for coat coloring traits. GWAS are presented for the proportion of daughters with red head (A1) and the proportion of daughters without spotting (B1) based on 15,182,131 imputed variants in 3062 Fleckvieh animals. In (B1) the largest effects emerge from eight SNPs summarized in Table 2 with MITF (P = 2.65e-58) and *KIT/KDR* (P = 2.46e-44) at the top. Together, *KIT/KDR* and MITF explained 36.25% of the residual variance of the trait in the studied population. A2 and B2 are the corresponding quantile-quantile plots. Shown in blue is the quantile-quantile plot resulting from removal of all SNPs in the region of significant genes listed in Table 2, for both traits. The shaded area is the 95% concentration band under the null hypothesis of no association. Panels C and D are detailed overviews of the associated regions on BTA6 and BTA22, respectively. Variants in the promoter (defined to encompass 1,000 bp upstream of the transcription start), in the untranslated regions (UTR) and in the amino-acid coding region are highlighted with different color. The red triangles indicate the genomic positions of KIT and MITF genes. doi:10.1371/journal.pgen.1004148.q005

three major alleles exist in cattle MC1R, the E^+ wild type, E^D dominant black locus, and e recessive red locus [35]. Since the red variant is fixed in Fleckvieh population, no [possible] association with coat spotting or red head can be traced through GWAS. This also explains why the locus is not very extreme when explored by the iHS statistic (see Figure S2A). We postulate that red hair has been under very recent selection (e.g., human driven selection during breed formation) resulting in fixation of the red variant in the Fleckvieh cattle, while simultaneously fixing alleles at nearby hitch-hiked loci. To validate this hypothesis, we calculated SNPspecific F_{ST} [62] for 1,173 SNPs on BTA18 which have been genotyped in 2,084 Holstein-Friesian (black and white coat color) and 2,539 Fleckvieh (red and white coat color) animals using 50K SNP arrays. This revealed a strong differentiation hit located at 14.42 Mb in immediate vicinity to MC1R, further supporting findings of selection analyses (Figure S2B). Therefore, our results exemplify how a historical selective sweep that underwent fixation can be localized by employing a population genetic approach.

We further conducted GWA studies for somatic cell count (SCC, Figure 6A) and body size (Figure S10) based on daughterderived phenotypes (estimated breeding values, EBVs) for these traits. In cattle breeding schemes, somatic cell count measured as the log number of somatic cells per *ml* of milk is widely used as an indicator for incidence of mastitis. The GWAS for SCC revealed strong association on BTA22 nearby the *LTF* gene encoding lactotransferrin (Table S5). Due to its antimicrobial and antiinflammatory activity [63], *LTF* is a strong candidate for mastitis resistance. The most significant SNP is located ~8 Kb upstream of the translation start of *LTF* and might affect expression of *LTF* during mammary gland infections [64]. Another QTL is located on BTA3 and the most significant SNP (rs386094483, P = 7.95e-11) is a Lysine-to-Arginine substitution in the *DC-STAMP* domain Table 2. A descriptive summary of GWA studies for coat color variation in Fleckvieh animals.

					Candidate gene	Proportion of EBV variation explained [%]
Chr	Position (bp)	Top-SNP (NCBI reference ID)	Minor allele frequency	P-Value		
Without spotting						
1	113,261,262	rs110220767	0.27	5.11e-14	MME	2.95
6	23,655,204	rs43461001	0.16	8.89e-11	NFKB1	1.26
6	72,085,585	rs135123206	0.07	2.46e-44	KIT/KDR	14.68
8	53,907,785	rs110041961	0.35	1.01e-9	GNA14	0.8
13	57,580,515	rs382817429	0.17	2.42e-36	EDN3	5.75
17	7,202,981	rs135947957	0.42	1.04e-9	LRBA	0.92
21	9,225,382	rs42261960	0.37	6.65e-17	ARRDC4	1.83
22	31,841,994	rs41642495	0.12	2.65e-58	MITF	21.57
Red Head						
6	71,404,818	rs137525659	0.10	6.8e-108	KIT	34.81

doi:10.1371/journal.pgen.1004148.t002

containing 1-encoding gene (*DCST1*, p.K113R, Chr3:15613949) (Figure 6C). *DCST1* plays an important role in the initiation of the immune system by antigen processing [65,66,67]. The affected amino acid is highly conserved among species and is a strong candidate causal mutation for the somatic cell count in cattle (Figure 6D).

The GWAS for body size revealed three QTLs. One example is the *PLAG1* region on BTA14 which is significantly associated with

body size (P = 1.5e-27). This region was already shown to affect growth-related traits in several species including cattle [68]. This QTL is also associated with the calving difficulties and an increased stillbirth rate in cattle, probably resulting from an enhanced fetal growth [69].

Table S5 presents a descriptive summary of the most significant SNPs and the proportion of EBV variance attributable to the five identified QTL regions. Altogether, the imputed variants explain



Figure 6. The visualization of the signals revealed by association analyses for SCC trait. Manhattan plot presents the association of 15'182'131 imputed SNPs with the "somatic cell count" trait in 3602 Fleckvieh animals (A). Panel B represents corresponding quantile-quantile plot. Shown in blue is the quantile-quantile plot resulting from excluding SNPs in the region of significant genes. The shaded area represents 95% concentration band under the null hypothesis of no association. Panel C displays the associated region on BTA3 with a higher resolution. Variants in the promoter (defined to encompass 1,000 bp upstream of the transcription start), in the untranslated regions (UTR) and in the amino-acid coding region are highlighted with different color. Panel D provides a multi species alignment of *DC-STAMP* domain containing 1 encoded by *DCST1* gene. doi:10.1371/journal.pgen.1004148.g006

81.35% and 80.73% of the variation of EBV, respectively for the SCC and body size traits. A schematic illustration of chromosomal contributions for the total variance of these traits are represented respectively, in Figures S11 and S12.

In contrast to regions associated with color variation QTLs, these regions show no overlap with selection candidates. One possible explanation is that most selection affecting body size and somatic cell count has affected alleles that are no longer segregating in this breed. This raises the question as to why the currently segregating alleles affecting these traits do not show strong signals of selection. Possibly, the alleles that are still segregating, even after intense artificial selection during domestication, may have negative pleiotropic effects preventing them from increasing in frequency in the population. Additionally, selection is likely to have affected standing variation. If the selected mutations were segregating on multiple different haplotypes before selection started, both the iHs and the CLR statistic may have limited power to detect selection.

We present the first comprehensive study for localizing signatures of past selection in cattle based on full re-sequencing data. 106 candidate regions were identified containing genes with biological functions involved in blood clotting, immune-defense functions, pigmentation pattern, sensory perceptions and neurobehavioral functioning. The detection of genes related to pigmentation is not surprising since a specific coat color pattern, such as red (in various shadings) coat with white head, is constitutive for the breed definition of Fleckvieh and therefore fundamental for the breed formation process. Using the same samples, we also performed a coat color GWA study, and show that there is a strong overlap between genes identified in the GWAS and in the selection scan. As demonstrated for the gene MC1R, selection signatures can be detected in regions where anthropogenic selection has fixed the desired allele and, consequently, GWAS fails. This illustrates the potential for population genetic techniques to identify genomic regions relating to phenotypes of importance to breeders. Comparing GWAS results for different traits also provides further evidence regarding the complexity of genetics underlying coat coloring in cattle.

Materials and Methods

Ethics statement

DNA needed for the study was previously extracted from commercial AI bull semen straws. No ethics statement is thus required.

Sequenced-based imputation

For the purpose of this study we used data from Jansen et al. [23]. Briefly, it consists low to medium coverage (\sim 7.4-fold) sequence of the entire genomes of 43 key and contemporary animals representing \sim 69% of the genetic diversity of the current German Fleckvieh population.

The sequence panel consisting of 15,182,131 SNPs with an average inter-marker space equal to 178 ± 115 bp was used for a two-step imputation using default setting in Beagle [70] and Minimac [71], respectively. Imputation started from a medium density panel (50K SNPs) bridged by a high density panel (700K SNPs) to the full sequences using 43 reference animals. We evaluated the accuracy of sequence-based imputation for chromosomes 5, 15 and 25 within the high density panel (700K SNPs). Genotypes for randomly selected 66% of the SNPs were retained, while genotypes for the remaining SNPs were masked to mimic missing genotypes. Those were imputed using Beagle and

Minimac (see above) based on sequence-derived genotypes of 43 re-sequenced animals. Imputation accuracy was assessed as the correlation between array-derived and imputed genotypes. This approach yielded high imputation accuracy for frequent alleles (e.g., MAF >5%) (Figure S13). However, the number of re-sequenced animals (n = 43) might not be sufficient for imputing low-frequency variants with a sophisticated accuracy, which agrees with a previous report in cattle [72].

The individual call-rate was >95% for all animals genotyped with SNP arrays. After quality control (call-rate per SNP >95%, minor allele frequency >0.5%, no significant deviation of the Hardy-Weinberg-Equilibrium (P>10-6), known chromosomal position), the medium and high density dataset comprised genotypes for 39,304 (n = 2,309) and 645,189 SNPs (n = 1,293), respectively.

Linkage disequilibrium

We quantified LD using the squared correlation coefficients (r^2) between pairs of SNPs. Evaluations of SNP to SNP pairwise r^2 were completed based on the panels of 15,000 SNPs randomly sampled across two segments of each 5 Mb on the chromosomes 5, 10, 15, 20 and 26. We then compared measures of r^2 to those of 50K and 700K bovine arrays across the corresponding chromosomes.

Detecting positive selection

Evidence of positive selection was investigated through multiple statistics: We performed the Composite Likelihood Ratio test (CLR) using information from allele frequencies to detect a completed sweep [11]. Briefly, CLR relies on identifying skews in the allele frequency spectrum toward excess of rare and frequent alleles. To infer ongoing sweeps we employed the integrated Haplotype Homozygosity Score (iHS) that explores the structure of haplotype and essentially indicates unusually long haplotypes carrying the ancestral and derived allele [12]. Single site values for iHS were averaged in non-overlapping windows of 40 Kb across the genome resulting in a total of 62'196 windows. Window size was adapted based on the extent of LD as discussed above (Figure 2). The variance of SNP statistics within 40 kb windows (i.e., Var|iHS| = 0.27) was significantly smaller than that among randomly selected SNPs (i.e., Var|iHS| = 0.35), confirming that the windows effectively grouped SNPs with more similar statistic values. Figure S14 visualizes the number of SNPs distributed across sliding windows. To produce comparable results of the composite likelihood ratio (CLR) test, which is a multi-locus statistic, the grid size was taken as 40 kb. This resulted in 62,788 local CLR values across the genome. The empirical P-values were generated by genome wide ranking of |iHS| and CLR values. The list of |iHS| and CLR scores for all windows are available in supporting material as Dataset S1 and Dataset S2, respectively. The iHS metric was calculated using the R package 'rehh' [73]. For estimating CLRs we used 'sweepfinder' [11] with a background allele frequency spectrum calibrated genome-wide. We further used custom programs to calculate the 'observed heterozygosity' (Het), which should be reduced in regions affected by a sweep [74], Tajima's D [8], Fay and Wu H statistic [9], and the number of Segregating Loci (nSL), a recently developed statistic related to iHS in Rasmus Nielsen's Lab. The metrics were estimated using different window size to explore the sensitivity to the choice of window. We annotated candidate genomic regions by aligning the positions to the bovine genome sequence assembly build 6.1, to reveal genes and ESTs located in the respective region.

Randomization test

To test if the overlap between a selection scan and a GWAS is significantly different from that expected at random, we performed permutation tests. We generated 10,000 simulated data sets by permuting p-values among either single SNPs or windows. For example, we calculated empirical p-values for windows based on |iHS|, then counted the number of instances where a GWAS association (p-value<10⁻⁶, Bonferroni corrected threshold) overlaps a high-scoring (top 1%) selection window. The observed value of the overlap statistic was then compared to the distribution of overlap statistics in the permuted data sets.

Sequenced-based association study

After imputation, genotypes of 15,182,131 SNPs in 3602 individuals were used for a GWAS for four different traits: the square-root transformed proportion of daughters without spotting and with red head and estimated breeding values for somatic cell counts and body size. EMMAX [75] was used to fit the model y = Xb + u + e, where y is the vector of phenotypes, b is the SNP effect, X is a design matrix of allele dosages for the imputed SNPs, u is the additive genetic effect $\sim N(0,\sigma_a^2,G)$, where σ_a^2 is the additive genetic variance, G is the realized genomic relationship matrix estimated using genotype information [76], and e is the random residual term.

Supporting Information

Figure S1 Evidence for the effects of MAF (A) and sample composition (B) on the strength of LD. Panel A compares LD from sequence (n = 43, red) and array (n = 1,293, green) data against SNP set with MAF >0.2 (n = 43, blue). In Panel B, LD curves from sequence (red) and array (green) data are compared against set of bovine 700K SNPs sub-selected from only sequences of only 43 individuals.

(TIF)

Figure S2 Selection for coat coloring. Panel A is detailed schematic illustration of the region harboring *MC1R* locus in Fleckvieh animals. The pattern of composite likelihood ratio (CLR), |iHS|, standardized heterozygosity (ZHet), Tajima D, number of Segregating Loci and Fay and Wu H values are depicted. The multi-locus CLRs are estimated in grid size = 5 Kb while other metrics are single SNP values accumulated in windows of 40 Kb and depicted in steps of 5 Kb. *MC1R* is located between 14,757,332 and 14,759,082 bp on BTA18. Panel B displays sliding 3-SNP windows of F_{ST} calculated between Holstein-Friesian (black coat color) and Fleckvieh (red coat color) animals based on 1173 SNPs on BTA18. The dashed line displays top 0.1% cutoff and the red triangle represents the position of *MC1R*. (TIF)

Figure S3 A detailed schematic illustration of the region harboring *NRG4* locus in Fleckvieh animals. The pattern of composite likelihood ratio (CLR), |iHS|, standardized heterozygosity (ZHet), Tajima D, number of Segregating Loci and Fay and Wu H values are depicted. The multi-locus CLRs are estimated in grid size = 5 Kb while other metrics are single SNP values accumulated in windows of 40 Kb and depicted in steps of 5 Kb. *NRG4* is located between 31,824,897 and 31,825,956 bp on BTA21. (TIF)

Figure S4 A detailed schematic illustration of the region harboring *Erbb4* locus in Fleckvieh animals. The pattern of composite likelihood ratio (CLR), [iHS], standardized heterozygosity (ZHet), Tajima D, number of Segregating Loci and Fay and Wu H values are depicted. The multi-locus CLRs are estimated in grid size = 5 Kb while other metrics are single SNP values accumulated in windows of 40 Kb and depicted in steps of 5 Kb. *Erbb4* is located between 99,660,620 and 100,632,794 bp on BTA2.

(TIF)

Figure S5 A detailed schematic illustration of the region harboring *TMEM132D* locus in Fleckvieh animals. The pattern of composite likelihood ratio (CLR), |iHS|, standardized heterozygosity (ZHet), Tajima D, number of Segregating Loci and Fay and Wu H values are depicted for the region harboring *TMEM132D* locus in Fleckvieh animals. The multi-locus CLRs are estimated in grid size = 5 Kb while other metrics are single SNP values accumulated in windows of 40 Kb and depicted in steps of 5 Kb. *TMEM132D* is located between 48,199,004 and 49,090,245 bp on BTA17.

(TIF)

Figure S6 A detailed schematic illustration of the region harboring *GRIK3* locus in Fleckvieh animals. The pattern of composite likelihood ratio (CLR), |iHS|, standardized heterozygosity (ZHet), Tajima D, number of Segregating Loci and Fay and Wu H values are depicted. The multi-locus CLRs are estimated in grid size = 5 Kb while other metrics are single SNP values accumulated in windows of 40 Kb and depicted in steps of 5 Kb. *GRIK3* is located between 109,422,556 and 109,667,304 bp on BTA3.

(TIF)

Figure S7 A detailed schematic illustration of the region harboring *OLIG1* locus in Fleckvieh animals. The pattern of composite likelihood ratio (CLR), |iHS|, standardized heterozygosity (ZHet), Tajima D, number of Segregating Loci and Fay and Wu H values are depicted. The multi-locus CLRs are estimated in grid size = 5 Kb while other metrics are single SNP values accumulated in windows of 40 Kb and depicted in steps of 5 Kb. *OLIG1* is located between 1,775,224 and 1,777,299 bp on BTA1.



Figure S8 A detailed schematic illustration of the regions harboring two candidate selective sweeps harboring *MAGEA13P*like and tescalcin-like loci. The pattern of composite likelihood ratio (CLR), |iHS|, standardized heterozygosity (ZHet), Tajima D, number of Segregating Loci and Fay and Wu H values are depicted. *MAGEA13P*-like and tescalcin-like loci are located, respectively on 42,269,085..42,270,099 bp and 42,686,629..42, 687,350 bp on BTA14. The multi-locus CLRs are estimated in grid size = 5 Kb while other metrics are single SNP values accumulated in windows of 40 Kb and depicted in steps of 5 Kb. (TIF)

Figure S9 Chromosomal partitioning of the phenotypic variance of the 'proportion of daughters without spotting'. The grey and blue bars indicate the fraction of phenotypic variance attributed to a particular chromosome and QTL region, respectively. The triangles represent the cumulative proportion of phenotypic variance attributable to the 30 chromosomes. (TIF)

Figure S10 Manhattan plot of association of 15'182'131 imputed variants with the body size in 3602 Fleckvieh animals (A). Panel B represents corresponding quantile-quantile plot. Shown in blue is the quantile-quantile plot resulting from excluding SNPs in the region of significant genes. The shaded

area represents 95% concentration band under the null hypothesis of no association. (TIF)

Figure S11 Chromosomal partitioning of the EBV variance of the 'somatic cell count' trait. The grey and blue bars indicate the fraction of phenotypic variance attributed to a particular chromosome and QTL region, respectively. The triangles represent the cumulative proportion of phenotypic variance attributable to the 30 chromosomes. (TIF)

Figure S12 Chromosomal partitioning of the EBV variance of the 'body size' trait. The grey and blue bars indicate the fraction of phenotypic variance attributed to a particular chromosome and QTL region, respectively. The triangles represent the cumulative proportion of phenotypic variance attributable to the 30 chromosomes.

(TIF)

Figure S13 Evaluation of imputation accuracy. Correlation between imputed and array-derived genotypes as a function of the minor allele frequency. Genotypes for randomly selected SNPs were set to missing and subsequently imputed based on sequence-derived genotypes of 43 re-sequenced animals. (TIF)

Figure S14 Distribution of the number of SNPs in 62'196 nonoverlapping windows along BTA1 to BTA29. (TIF)

Table S1 Comparison of fraction (%) of marker pairs with different r2 levels as a function of distance (kb). (DOCX)

References

- Bradley DG, Cunningham EP (1999) Genetic aspects of domestication. In: Fries R, Ruvinsky A, Eds. The Genetics of Cattle. pp 15–31.
- Bollongino R, Burger J, Powell A, Mashkour M, Vigne J-D, et al. (2012) Modern Taurine Cattle Descended from Small Number of Near-Eastern Founders. Mol Biol Evol 29: 2101–2104. doi:10.1093/molbev/mss092.
- Loftus RT, MacHugh DE, Bradley DG, Sharp PM, Cunningham P (1994) Evidence for two independent domestications of cattle. Proc Natl Acad Sci U S A 91: 2757–2761.
- Troy CS, MacHugh DE, Bailey JF, Magee DA, Loftus RT, et al. (2001) Genetic evidence for Near-Eastern origins of European cattle. Nature 410: 1088–1091. doi:10.1038/35074088.
- FAO (2007) The state of the world's animal genetics resources for food and agriculture. ftp://ftp.fao.org/docrep/fao/010/a1250e/a1250e02.pdf
- Zeder MA."Pathways to Animal Domestication (Melinda A. Zeder, 2012)," in BoneCommons, Item #11838, http://alexandriaarchive.org/bonecommons/ items/show/1838 (accessed April 5, 2013).
- The Bovine HapMap Consortium, Gibbs RA, Taylor JF, Van Tassell CP, Barendse W, et al. (2009) Genome-Wide Survey of SNP Variation Uncovers the Genetic Structure of Cattle Breeds. Science 324: 528–532. doi:10.1126/ science.1167936.
- Tajima F (1989) Statistical method for testing the neutral mutation hypothesis by DNA polymorphism. Genetics 123: 585–595.
- Fay JC, Wu C-I (2000) Hitchhiking Under Positive Darwinian Selection. Genetics 155: 1405–1413.
- Sabeti PC, Reich DE, Higgins JM, Levine HZP, Richter DJ, et al. (2002) Detecting recent positive selection in the human genome from haplotype structure. Nature 419: 832–837. doi:10.1038/nature01140.
- Nielsen R, Williamson S, Kim Y, Hubisz MJ, Clark AG, et al. (2005) Genomic scans for selective sweeps using SNP data. Genome Res 15: 1566–1575. doi:10.1101/gr.4252305.
- Voight BF, Kudaravalli S, Wen X, Pritchard JK (2006) A Map of Recent Positive Selection in the Human Genome. PLoS Biology 4: e72. doi:10.1371/ journal.pbio.0040072.
- Rubin C-J, Zody MC, Eriksson J, Meadows JRS, Sherwood E, et al. (2010) Whole-genome resequencing reveals loci under selection during chicken domestication. Nature 464: 587–591. doi:10.1038/nature08832.
- Nielsen R (2000) Estimation of Population Parameters and Recombination Rates From Single Nucleotide Polymorphisms. Genetics 154: 931–942.

Table S2 Summary statistics of testing the difference between sequence vs., array based LD.

 (DOCX)

 Table S3
 Candidate regions identified by iHS analysis.

 (DOCX)
 (DOCX)

 Table S4
 Candidate regions identified by CLR analysis.

 (DOCX)
 (DOCX)

Table S5Results of GWA studies based on estimated breedingvalues for Somatic cell count and body size in Fleckvieh animals.(DOCX)

Dataset S1 Genome wide iHS scores. (CSV)

Dataset S2 Genome wide CLR scores. (CSV)

Acknowledgments

We acknowledge the Bavarian State Research Center (LfL) for providing phenotype data. Part of this work was carried out while SQ was visiting The University of California at Berkeley as a visiting scientist. We thank Filipe G. Vieira, Matteo Fumagalli, Gaston Sanchez, Kirk Lohmueller and Anna Ferrer-Admetlla for technical supports and their useful discussions.

Author Contributions

Conceived and designed the experiments: SQ HS RN. Performed the experiments: TMS SJ RF. Analyzed the data: SQ HP SJ MS. Contributed reagents/materials/analysis tools: RF TMS SJ. Wrote the paper: SQ RN HS.

- Kuhner MK, Beerli P, Yamato J, Felsenstein J (2000) Usefulness of Single Nucleotide Polymorphism Data for Estimating Population Parameters. Genetics 156: 439–447.
- Wakeley J, Nielsen R, Liu-Cordero SN, Ardlie K (2001) The Discovery of Single-Nucleotide Polymorphisms-and Inferences about Human Demographic History. Am J Hum Genet 69: 1332–1347. doi:10.1086/324521.
- Lynn DJ, Freeman AR, Murray C, Bradley DG (2005) A Genomics Approach to the Detection of Positive Selection in Cattle: Genetics 170: 1189–1196. doi:10.1534/genetics.104.039040.
- MacEachern S, Hayes B, McEwan J, Goddard M (2009) An examination of positive selection and changing effective population size in Angus and Holstein cattle populations (Bos taurus) using a high density SNP genotyping platform and the contribution of ancient polymorphism to genomic diversity in Domestic cattle. BMC Genomics 10: 181. doi:10.1186/1471-2164-10-181.
- Flori L, Fritz S, Jaffrézic F, Boussaha M, Gut I, et al. (2009) The Genome Response to Artificial Selection: A Case Study in Dairy Cattle. PLoS ONE 4: e6595. doi:10.1371/journal.pone.0006595.
- Stella A, Ajmone-Marsan P, Lazzari B, Boettcher P (2010) Identification of Selection Signatures in Cattle Breeds Selected for Dairy Production. Genetics 185: 1451–1461. doi:10.1534/genetics.110.116111.
- Qanbari S, Pimentel ECG, Tetens J, Thaller G, Lichtner P, et al. (2010b) A genome-wide scan for signatures of recent selection in Holstein cattle. Anim Genet 41: 377–389. doi:10.1111/j.1365-2052.2009.02016.x.
- Qanbari S, Gianola D, Hayes B, Schenkel F, Miller S, et al. (2011) Application of site and haplotype-frequency based approaches for detecting selection signatures in cattle. BMC Genomics 12: 318. doi:10.1186/1471-2164-12-318.
- Jansen S, Aigner B, Pausch H, Wysocki M, Eck S, et al. (2013) Assessment of the genomic variation in a cattle population by re-sequencing of key animals at low to medium coverage. BMC Genomics 14: 446. doi:10.1186/1471-2164-14-446.
- Sargolzaei M, Schenkel FS, Jansen GB, Schaeffer LR (2008) Extent of linkage disequilibrium in Holstein cattle in North America. J Dairy Sci 91: 2106–2117. doi:10.3168/jds.2007-0553.
- Qanbari S, Pimentel ECG, Tetens J, Thaller G, Lichtner P, et al. (2010a) The pattern of linkage disequilibrium in German Holstein cattle. Anim Genet 41: 346–356. doi:10.1111/j.1365-2052.2009.02011.x.
- Boyko AR, Williamson SH, Indap AR, Degenhardt JD, Hernandez RD, et al. (2008) Assessing the Evolutionary Impact of Amino Acid Mutations in the Human Genome. PLoS Genet 4: e1000083. doi:10.1371/journal.pgen.1000083.

- Li Y, Vinckenbosch N, Tian G, Huerta-Sanchez E, Jiang T, et al. (2010) Resequencing of 200 human exomes identifies an excess of low-frequency nonsynonymous coding variants. Nat Genet 42: 969–972. doi:10.1038/ng.680.
- Pritchard JK, Przeworski M (2001) Linkage Disequilibrium in Humans: Models and Data. Am J Hum Genet 69: 1–14.
- Huang DW, Sherman BT, Lempicki RA (2008) Systematic and integrative analysis of large gene lists using DAVID bioinformatics resources. Nat Protocols 4: 44–57. doi:10.1038/nprot.2008.211.
- Hayes BJ, Pryce J, Chamberlain AJ, Bowman PJ, Goddard ME (2010) Genetic Architecture of Complex Traits and Accuracy of Genomic Prediction: Coat Colour, Milk-Fat Percentage, and Type in Holstein Cattle as Contrasting Model Traits. PLoS Genet 6: e1001139. doi:10.1371/journal.pgen.1001139.
- Haase B, Brooks SA, Tozaki T, Burger D, Poncet P-A, et al. (2009) Seven novel KIT mutations in horses with white coat colour phenotypes. Animal Genetics 40: 623–629. doi:10.1111/j.1365-2052.2009.01893.x.
- Rubin C-J, Megens H-J, Barrio AM, Maqbool K, Sayyab S, et al. (2012) Strong signatures of selection in the domestic pig genome. PNAS 109: 19529–19536. doi:10.1073/pnas.1217149109.
- Baxter LL, Hou L, Loftus SK, Pavan WJ (2004) Spotlight on Spotted Mice: A Review of White Spotting Mouse Mutants and Associated Human Pigmentation Disorders. Pigment Cell Research 17: 215–224. doi:10.1111/j.1600-0749. 2004.00147.x.
- Hou L, Panthier JJ, Arnheiter H (2000) Signaling and transcriptional regulation in the neural crest-derived melanocyte lineage: interactions between KIT and MITF. Development 127: 5379–5389.
- Klungland H, Vage DI, Gomez-Raya L, Adalsteinsson S, Lien S (1995) The role of melanocyte-stimulating hormone (MSH) receptor in bovine coat color determination. Mammalian Genome 6: 636–639. doi:10.1007/BF00352371.
- Choi J, Young JAT, Callaway EM (2010) Selective viral vector transduction of ErbB4 expressing cortical interneurons in vivo with a viral receptor–ligand bridge protein. PNAS 107: 16703–16708. doi:10.1073/pnas.1006233107.
- Pickrell JK, Coop G, Novembre J, Kudaravalli S, Li JZ, et al. (2009) Signals of recent positive selection in a worldwide sample of human populations. Genome Res 19: 826–837. doi:10.1101/gr.087577.108.
- Stefansson H, Petursson H, Sigurdsson E, Steinthorsdottir V, Bjornsdottir S, et al. (2002) Neuregulin 1 and Susceptibility to Schizophrenia. The American Journal of Human Genetics 71: 877–892. doi:10.1086/342734.
- Mei L, Xiong W-C (2008) Neuregulin 1 in neural development, synaptic plasticity and schizophrenia. Nat Rev Neurosci 9: 437–452. doi:10.1038/ nrn2392.
- Petukhova L, Duvic M, Hordinsky M, Norris D, Price V, et al. (2010) Genomewide association study in alopecia areata implicates both innate and adaptive immunity. Nature 466: 113–117. doi:10.1038/nature09114.
- Erhardt A, Akula N, Schumacher J, Czamara D, Karbalai N, et al. (2012) Replication and meta-analysis of TMEM132D gene variants in panic disorder. Transl Psychiatry 2: e156. doi:10.1038/tp.2012.85.
- Nyegaard M, Demontis D, Foldager L, Hedemand A, Flint TJ, et al. (2010) CACNA1C (rs1006737) is associated with schizophrenia. Mol Psychiatry 15: 119–121. doi:10.1038/mp.2009.69.
- Brown SM, Clapcote SJ, Millar JK, Torrance HS, Anderson SM, et al. (2011) Synaptic modulators Nrxn1 and Nrxn3 are disregulated in a Disc1 mouse model of schizophrenia. Mol Psychiatry 16: 585–587. doi:10.1038/mp.2010.134.
- Yu L, Arbez N, Nucifora LG, Sell GL, Delisi LE, et al. (2013) A mutation in NPAS3 segregates with mental illness in a small family. Mol Psychiatry. doi:10.1038/mp.2012.192.
- Minelli A, Scassellati C, Bonvicini C, Perez J, Gennarelli M (2009) An association of GRIK3 Ser310Ala functional polymorphism with personality traits. Neuropsychobiology 59: 28–33. doi:10.1159/000202827.
- Barak T, Kwan KY, Louvi A, Demirbilek V, Saygi S, et al. (2011) Recessive LAMC3 mutations cause malformations of occipital cortical development. Nat Genet 43: 590–594. doi:10.1038/ng.836.
- Ache BW, Young JM (2005) Olfaction: Diverse Species, Conserved Principles. Neuron 48: 417–430. doi:10.1016/j.neuron.2005.10.022.
- Moreno-Estrada A, Casals F, Ramírez-Soriano A, Oliva B, Calafell F, et al. (2008) Signatures of Selection in the Human Olfactory Receptor OR511 Gene. Mol Biol Evol 25: 144–154. doi:10.1093/molbev/msm240.
- Chen R, Irwin DM, Zhang Y-P (2012) Differences in Selection Drive Olfactory Receptor Genes in Different Directions in Dogs and Wolf. Mol Biol Evol 29: 3475–3484. doi:10.1093/molbev/mss153.
- Groenen MAM, Archibald AL, Uenishi H, Tuggle CK, Takeuchi Y, et al. (2012) Analyses of pig genomes provide insight into porcine demography and evolution. Nature 491: 393–398. doi:10.1038/nature11622.
- Elsik CG, Tellam RL, Worley KC (2009) The Genome Sequence of Taurine Cattle: A Window to Ruminant Biology and Evolution. Science 324: 522–528. doi:10.1126/science.1169588.
- Dong D, Jones G, Zhang S (2009) Dynamic evolution of bitter taste receptor genes in vertebrates. BMC Evolutionary Biology 9: 12. doi:10.1186/1471-2148-9-12.
- Luca F, Perry GH, Di Rienzo A (2010) Evolutionary Adaptations to Dietary Changes. Annual Review of Nutrition 30: 291–314. doi:10.1146/annurev-nutr-080508-141048.
- Kosiol C, Vinař T, da Fonseca RR, Hubisz MJ, Bustamante CD, et al. (2008) Patterns of Positive Selection in Six Mammalian Genomes. PLoS Genet 4: e1000144. doi:10.1371/journal.pgen.1000144.

- 55. Athanasiadis G, Esteban E, Gayà -Vidal M, Dugoujon J-M, Moschonas N, et al. (2010) Different Evolutionary Histories of the Coagulation Factor VII Gene in Human Populations?: Evolutionary Patterns of the F7 Gene. Annals of Human Genetics 74: 34–45. doi:10.1111/j.1469-1809.2009.00557.x.
- Utsunomiya YT, Pérez O'Brien AM, Sonstegard TS, Van Tassell CP, do Carmo AS, et al. (2013) Detecting Loci under Recent Positive Selection in Dairy and Beef Cattle by Combining Different Genome-Wide Scan Methods. PLoS ONE 8: e64280. doi:10.1371/journal.pone.0064280.
- Saldana-Caboverde A, Kos L (2010) Roles of endothelin signaling in melanocyte development and melanoma. Pigment Cell & Melanoma Research 23: 160–170. doi:10.1111/j.1755-148X.2010.00678.x.
- Aberdam E, Auberger P, Ortonne J-P, Ballotti R (2000) Neprilysin, a Novel Target for Ultraviolet B Regulation of Melanogenesis Via Melanocortins. Journal of Investigative Dermatology 115: 381–387. doi:10.1046/j.1523-1747. 2000.00075.x.
- Yang J, Lee SH, Goddard ME, Visscher PM (2011) GCTA: A Tool for Genome-wide Complex Trait Analysis. The American Journal of Human Genetics 88: 76–82. doi:10.1016/j.ajhg.2010.11.011.
- Pausch H, Wang X, Jung S, Krogmeier D, Edel C, et al. (2012) Identification of QTL for UV-Protective Eye Area Pigmentation in Cattle by Progeny Phenotyping and Genome-Wide Association Analysis. PLoS ONE 7: e36346. doi:10.1371/journal.pone.0036346.
- Kijas JM, Moller M, Plastow G, Andersson L (2001) A frameshift mutation in MC1R and a high frequency of somatic reversions cause black spotting in pigs. Genetics 158: 779–785.
- Akey JM, Zhang G, Zhang K, Jin L, Shriver MD (2002) Interrogating a highdensity SNP map for signatures of natural selection. Genome Res 12: 1805– 1814. doi:10.1101/gr.631202.
- Nuijens JH, Berkel PHC van, Schanbacher FL (1996) Structure and biological actions of lactoferrin. J Mammary Gland Biol Neoplasia 1: 285–295. doi:10.1007/BF02018081.
- Jensen K, Günther J, Talbot R, Petzl W, Zerbe H, et al. (2013) Escherichia coliand Staphylococcus aureus-induced mastitis differentially modulate transcriptional responses in neighbouring uninfected bovine mammary gland quarters. BMC Genomics 14: 36. doi:10.1186/1471-2164-14-36.
- Hartgers FC, Vissers JL, Looman MW, van Zoelen C, Huffine C, et al. (2000) DC-STAMP, a novel multimembrane-spanning molecule preferentially expressed by dendritic cells. Eur J Immunol 30: 3585–3590. doi:10.1002/1521-4141(200012)30:12<3585::AID-IMMU3585>3.0.CO;2-Y.
- Sawatani Y, Miyamoto T, Nagai S, Maruya M, Imai J, et al. (2008) The role of DC-STAMP in maintenance of immune tolerance through regulation of dendritic cell function. Int Immunol 20: 1259–1268. doi:10.1093/intimm/ dxn082.
- Sanecka A, Ansems M, Prosser AC, Danielski K, Warner K, et al. (2011) DC-STAMP knock-down deregulates cytokine production and T-cell stimulatory capacity of LPS-matured dendritic cells. BMC Immunol 12: 57. doi:10.1186/ 1471-2172-12-57.
- Karim L, Takeda H, Lin L, Druet T, Arias JAC, et al. (2011) Variants modulating the expression of a chromosome domain encompassing PLAG1 influence bovine stature. Nat Genet 43: 405–413. doi:10.1038/ng.814.
- Pausch H, Flisikowski K, Jung S, Emmerling R, Edel C, et al. (2011) Genomewide association study identifies two major loci affecting calving ease and growth-related traits in cattle. Genetics 187: 289–297. doi: 10.1534/genetics. 110.124057
- Browning BL, Browning SR (2009) A Unified Approach to Genotype Imputation and Haplotype-Phase Inference for Large Data Sets of Trios and Unrelated Individuals. The American Journal of Human Genetics 84: 210–223. doi:10.1016/j.ajhg.2009.01.005.
- Howie B, Fuchsberger C, Stephens M, Marchini J, Abecasis GR (2012) Fast and accurate genotype imputation in genome-wide association studies through prephasing. Nat Genet 44: 955–959. doi:10.1038/ng.2354.
- Pausch H, Aigner B, Emmerling R, Edel C, Götz K-U, et al. (2013) Imputation of high-density genotypes in the Fleckvieh cattle population. Genetics Selection Evolution 45: 3. doi:10.1186/1297-9686-45-3.
- Gautier M, Vitalis R (2012) rehh: an R package to detect footprints of selection in genome-wide SNP data from haplotype structure. Bioinformatics 28: 1176– 1177. doi:10.1093/bioinformatics/bts115.
- Smith JM, Haigh J (1974) The hitch-hiking effect of a favourable gene. Genetics Research 23: 23–35. doi:10.1017/S0016672300014634.
- Kang HM, Sul JH, Service SK, Zaiden NA, Kong S, et al. (2010) Variance component model to account for sample structure in genome-wide association studies. Nat Genet 42: 348–354. doi:10.1038/ng.548.
- Hayes BJ, Bowman PJ, Chamberlain AC, Verbyla K, Goddard ME (2009) Accuracy of genomic breeding values in multi-breed dairy cattle populations. Genetics Selection Evolution 41: 51. doi:10.1186/1297-9686-41-51.
- Hiraoka S, Furuichi T, Nishimura G, Shibata S, Yanagishita M, et al. (2007) Nucleotide-sugar transporter SLC35D1 is critical to chondroitin sulfate synthesis in cartilage and skeletal development in mouse and human. Nat Med 13: 1363– 1367. doi:10.1038/nm1655.
- Zhu H, Shah S, Shyh-Chang N, Shinoda G, Einhorn WS, et al. (2010) Lin28a transgenic mice manifest size and puberty phenotypes identified in human genetic association studies. Nat Genet 42: 626–630. doi:10.1038/ng.593.